Author's response to reviews

Title: Understanding comprehensibility and acceptance of an evidence-based consumer information brochure on fall prevention in old age: a focus group study

Authors:

Sabine Lins (Sabine.Lins@gmx.de)
Andrea Icks (Andrea.Icks@uni-duesseldorf.de)
Gabriele Meyer (Gabriele.Meyer@uni-wh.de)

Version: 6 Date: 8 May 2011

Author's response to reviews:

Dear Editor,

Thank you for giving us the opportunity to further revise our paper.

We made all suggested changes and hope that the manuscript will now be prepared for publication.

Thank you very much for your patience.

With kind regards,

Gabriele Meyer